%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%                File: OpEx_temp.tex                  %
%                  Date: Sept. 2, 2009                %
%                                                     %
%           LaTeX template file for use with          %
%           OSA's journal Optics Express              %
%                                                     %
%  send comments to Jennifer Mayfield, jmayfi@osa.org %
%                                                     %
% This file requires style file, opex3.sty, under     %
%              the LaTeX article class                %
%                                                     %
%   \documentclass[10pt,letterpaper]{article}         %
%   \usepackage{opex3}                                %
%                                                     %
% Note that our online submission system does not     %
% currently process PDFLaTeX; if PDFLaTeX must be     %
% used, pls. contact OpEx staff, and we will process  %
% manually                                            %
%                                                     %
%                                                     %
%       (c) 2009 Optical Society of America           %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%% preamble %%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass[10pt,letterpaper]{article}
\usepackage{opex3,epstopdf}
\newcommand{\ket}[1]{\mid #1 \rangle}
 %\usepackage{ae} %%for Computer Modern fonts

%%%%%%%%%%%%%%%%%%%%%%% begin %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{99}

\bibitem{jennewein00} Thomas Jennewein, Christoph Simon, Gregor Weihs, Harald Weinfurter, and Anton Zeilinger, \emph{Quantum Cryptography with Entangled Photons}, Phys. Rev. Lett. 84, 4729 (2000).

\bibitem{knill01}  E. Knill, R Laflamme, G milburn, \emph{A scheme for efficient quantum computation with linear optics}, Nature 409, 46 (2001).

\bibitem{migdell99}  A. Migdell, \emph{Correlated-photon metrology without absolute standards}, Physics Today Jan 41-46 (1999).

\bibitem{nam08} Adriana E. Lita, Aaron J. Miller, and Sae Woo Nam, /emph{Counting near-infrared single-photons with 95% efficiency}, Opt. Express 16, 3032 (2008) 

\bibitem{kwiat94} Paul G. Kwiat, Philippe H. Eberhard, Aephraim M. Steinberg, and Raymond Y. Chiao, \emph{Proposal for a loophole-free Bell inequality experiment}, Phys. Rev. A 49, 3209–3220 (1994). 

\bibitem{Fiorentino04} Marco Fiorentino, Gaetan Messin, Christopher E. Kuklewicz, Franco N. C. Wong, and Jeffrey H. Shapiro , \emph{Generation of  ultrabright tunable polarization entanglement without spatial, spectral, or temporal  constraints}, Phys. Rev. A 69, 041801(R) (2004).



\bibitem{koing05} F. Konig, EJ Mason, FNC Wong, and MA Albota, \emph{Efficient and spectrally bright source of polarization entangled photons}, Phys. Rev. A 71, 033805 (2005). 

\bibitem{zeilinger07} M. Hentschel, A poppe and A. Zeilinger, \emph{Three-color Sagnac source of polarization-entangled photon pairs} Opt. Express 15, 15377 (2007).
 
\bibitem{eberhard93} P.H. Eberhard, \emph{Background level and counter efficiencies required for a loophole-free Einstein-Podolsky-Rosen experiment} Phys Rev. A 47, R747 (1993)



\bibitem{schuck10} C. Schuck, F. Rohde, N. Piro, M. Almendros, J. Huwer, M. W. Mitchell, M. Hennrich, A. Haase, F. Dubin, and J. Eschner, \emph{Resonant interaction of a single atom with single photons from a down-conversion source}, Phys. Rev. A 81, 011802(R) (2010).
 
\bibitem{bennink10} Ryan S. Bennink, \emph{Optimal collinear Gaussian beams for spontaneous parametric down-conversion} Phys. Rev. A 81, 053805 (2010).

\bibitem{zeilinger13} M. Giustina, A. Mech, S. Ramelow, B. Wittmann, J. Kofler, J. Beyer,	A. Lita,	 B. Calkins, T. Gerrits, S. W. Nam, R. Ursin, A. Zeilinger,\emph{Bell violation using entangled photons without the fair-sampling assumption}     Nature 497, 227–230  (2013).
\bibitem{migdall13}M. D. C. Pereira, F. E. Becerra, B. L. Glebov, J. Fan, S. W. Nam, A. Migdall, \emph{Demonstrating highly symmetric single-mode, single-photon heralding efficiency in spontaneous parametric downconversion} Optics Letters, 38, 10, 1609-1611 (2013).
\bibitem{kwiat13}B.G. Christensen, K.T. McCusker, J.B. Altepeter, B. Calkins, T. Gerrits, A.E. Lita, A. Miller, L.K. Shalm, Y. Zhang, S.W. Nam, N. Brunner, C.C.W. Lim, N. Gisin, P.G. Kwiat, \emph{Detection-Loophole-Free Test of Quantum Nonlocality, and Applications} PRL 111, 130406 (2013).
}
\end{thebibliography}


%%%%%%%%%%%%%%%%%% title page information %%%%%%%%%%%%%%%%%%
\title{A high heralding efficiency source of polarization entangled photon pairs}

\author{authors}

\address{Center for Quantum Technologies, National University of Singapore, \\ Singapore, 117543}

\email{siddarth@nus.edu.sg} %% email address is required

% \homepage{http:...} %% author's URL, if desired

%%%%%%%%%%%%%%%%%%% abstract and OCIS codes %%%%%%%%%%%%%%%%
%% [use \begin{abstract*}...\end{abstract*} if exempt from copyright]

\begin{abstract}
%We present a PPKTP based source for polarization-entangled photon pairs with an efficiency for generating photon pairs into singlemode fibers that should be high enough for a loophole free test of a Bell inequality. By optimising the focusing conditions and mode overlap between the pump and downconverted modes we observe uncorrected heralding efficiencies of $>$39\,\%\
\end{abstract}

\ocis{(270.0270) Quantum Optics; (190.4410) Nonlinear Optics, parametric processes} % REPLACE WITH CORRECT OCIS CODES FOR YOUR ARTICLE

%%%%%%%%%%%%%%%%%%%%%%% References %%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%  body  %%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Introduction}

Sources of polarization entangled photon pairs or heralded photons are a fundamental resource for a wide range of fields like quantum communication\cite{jennewein00}, computation\cite{knill01} and metrology\cite{migdell99}. The underlying protocols behind these applications often require a complete detection of all photons to outperform their classical counterparts. While photodetectors have come close to unit detection efficiency\cite{nam08}, photon pair sources seem to be the current bottleneck in applications requiring a high efficiency. Likewise, the efficient transfer of information from photons to microscopic systems like single atoms or molecules requires an optical bandwidth that is much smaller than what most current photon pair sources can provide.
 In this paper, we present a photon pair source which addresses these issues. A pump laser beam ( $\lambda$= 405 nm) is focused into a type II PPKTP crystal, where it undergoes spontaneous parametric down-conversion into signal and idler modes which are collinear with the pump mode, maximizing the mode overlap between the target modes. 



\section{The source}
\begin{figure}[htbp]
\centering\includegraphics[width=7cm]{data/Sagnac_with_Calcite}
\caption{\label{fig:setup}Experimental setup, showing independent alignment degrees of freedom for the pump and downconvertedmodes). {\bf This figure needs to be edited. The pump lenses are now a telescope before the the pump pbs and the tes detectors need to be changed into apds}}
\end{figure}
\subsection{Source components}
\subsubsection{Crystal description name and length}
A 25\,mm Periodically Polled potassium Titanyl Phosphate ($KTiOPO_{4}$, PPKTP) crystal
\subsubsection{Pump description}
is pumped from both directions by 405nm light. This pump light is generated by an Ondax 405nm laser diode (with a bandwidth of 160\,MHz (datasheet)) which is mode filtered by using a singlemode fiber. A blue glass filter is used to eliminate any IR fluorescence from the pump fiber. The correct polarization is maintained by using a glan taylor polarizer. After passing through a beam splitter and reflecting off several mirrors the two pump modes are focused into the crystal. The waist of the pump modes can be varied by changing a
pair of lenses (see Fig.\,\ref{fig:setup}.).  
\subsubsection{Downconv style}
The crystal its self is a type $II$ donconversion crystal which means that the pair of photons produced will consist of a horizontally and a vertically polarized photon. The geometry used in our setup is a collinear one. 
%Thus when the crystal is pumped from both sides the downconverted modes will reflect off the dichroic mirrors and inteferrometrically recombine at the right PBS. Hence entanglement is produced due to the indistinguishably of the two possible downconversion paths. 
The downconverted modes are then coupled into single mode fibers after reflection off of one more dichroic mirror and passing through an interference filter; both of which serve to remove any residual pump light. 
\subsubsection{waists}
The downconverted modes from the crystal are focused into singlemode 780-HP fibers by a lens in each arm placed just after the PBS and aspheric lenses in the fiber couplers. We experimentally optimized the focusing of the pump and collection modes for the maximum efficiency (pairs to singles ratio). We use a pump waist of $265\,\mu$m and collection waists of $160\,\mu$m all of which are centered in the crystal.
\subsubsection{Degenaracy/temp tuning}
\begin{figure}[htbp] 
\centering\includegraphics[width=7cm, angle=-90]{data/WavelengthVsTEmperature_Lpump}
\caption{\label{fig:degen}Degeneracy of the down converted wavelengths is obtained at a $31.5\,^{\circ}\mathrm{C}$}
\end{figure}
The PPKTP crystal is placed in an oven and temperature stabilized to 10mK. By tuning the temperature of the crystal we can change the wavelength of the signal and idler. We measure the wavelength using an home built grating spectrometer connected to a single photon detector (Silicon Avalanche Photo Diode (Si APD)).   We find the degeneracy temperature to be $31.5\,^\circ$C (Fig.\,\ref{fig:degen}.).
\subsubsection{Bandwidth}
  We also measure the bandwidth of the downconverted light by temperature tuning a 0.1mm thick etalon. The bandwidth of the signal and idler is $128\pm4$\,GHz.
Using a michelson interferometer we can measure the bandwidth of the downconverted light see Fig.\,\ref{fig:fringe}
 {\bf This is by rotating an etalon, yet to take the temp tuning data}
\begin{figure}[htbp] 
\centering\includegraphics[width=7cm, angle=-90]{data/Pairs_Rcollec_Lpump_temp31p45C}
\caption{\label{fig:fringe}{\bf this is a place holder only. the proper data is on its way}}
\end{figure}



%This process can be done repeatedly for different pump beam waists. However, the compact geometry of the sagnac loop does not allow us to easily test a wide variety of focusing conditions. Instead we use a simple linear single pass setup to test the various focusing parameters the results of which are seen in {\bf Fig}.\,2.
\subsection{Entanglement}
\subsubsection{, how to get}
To obtain polarization entangled photons, we pump the crystal from both directions and interferometrically combine the two downconverted paths in a Sagnac interferometer\cite{kwiat94}.
\subsubsection{Phase stability}
 The original implementation\cite{Fiorentino04} used the same mirrors and PBS for both the pump and target modes, hence auto compensating for path length differences between pump and downconverted modes.  Our modified geometry (see Fig. 1.) allows for a decoupling of alignment degrees of freedom and also for independent optimization of  optical components. However,  this design now requires interferometric stability since the auto compensation feature of the Sagnac geometry is lost. We introduce a thin glass plate into one of the pump arms which we can rotate to periodically correct the phase of one pump direction with respect to the other\cite{Fiorentino04}{\bf not sure about this citation need to check}.  
\subsubsection{Alignment}
We optimize the mode overlap by alignment. To this end we first fix the alignment of the inner sagnac loop (the red path in Fig.\,1.) to see the zeroth fringe of this sagnac interferometer. This also fixes the alignment of the collection optics. We then align the pump using our specially folded sagnac geometry, which introduces an extra mirror, such that the downconverted light is collected with optimal efficiency. 
\subsection{Focus optimization}
For the mode overlap optimization we must also adjust the focusing of the aspheric lenses used to couple light to the singlemode fibers. It is important to center the waists in the crystal to preserve the symmetry of the pump and collection modes form either direction. 
\subsubsection{Measurement pol, aux dets}
In each collection arm we introduced a PBS and a motorized half wave plate. Together they serve as the measurement polarizers. The other output port of both of these PBS cubes is coupled into an auxiliary set of collection fibers. We connect these fibers to a pair of auxiliary detectors. We use these detectors to periodically readjust the phase difference between the pump modes ($\phi$). This leaves the primary collection fibers free to connect to high efficiency single photon detectors (TES). 
\subsubsection{State generation}
To generate the state
\begin{equation}
\ket{\psi} = sin (\theta) \ket {HV} + e^{i\phi} cos(\theta) \ket{VH}
\end{equation}
We introduce a motorized half wave plate before the pump PBS. This allows us to control the ratio of pump powers in each arm ($\theta$) and a glass cover slip in one pump arm which we can tilt to change the phase of one pump arm with respect to the other. This phase plate allows us to control $\phi$.
\subsubsection{Visibility}
For a maximally entangled state we can measure the polarization correlation in the $45\,^\circ$ basis by rotating one of the half wave plates in the collection arm (Fig.\,3.). The visibility obtained is $99.4\pm0.2\,\%$. 
\begin{figure}[htbp]
\centering\includegraphics[width=7cm, angle = -90]{data/vis_45_L300_R300_P0_ScanRhwp_RightHWP_3_inttime_800_20140111_1830}
\caption{\label{fig:vis}Visibility in the 45 deg basis is $99.4\pm0.2\,\%$}
\end{figure}
\subsubsection{Locking}
Using the signal on the auxiliary detectors, the phase between the two pump modes and consequently the phase of the two downconverted modes (HV and VH) is periodically readjusted. This ensures that the source produces a constant state $\ket{psi}$. 
\subsubsection{Stability}
Using a maximally entangled state we measure the polarization correlation visibility repeatedly for more than 6 hours and find it to be stable. The average visibility is $99.37\pm0.15\,\%$(Fig.\,\ref{fig:vis},Fig.\,\ref{fig:stability} .).
\begin{figure}[htbp]
\centering\includegraphics[width=7cm, angle = -90]{data/VisibilitySatbility_Processed}
\caption{\label{fig:stability}Visibility Stability in the $\pm45\,^{\circ}$ basis. By periodically readjusting the phase difference between the two pump modes we are able to maintain a stable visibility of $99.37\pm0.15\,\%$}
\end{figure}
\subsubsection{Non maximally}
When set to produce a particular non maximally entangled state we can perform an in-plane tomography for linear polarization states.  The reconstructed density matrices have a fidelity of $99.3\pm0.1\,\%$. This is of particular importance for experiments like device independent random number generation and loophole free violation of a Bell inequality with non maximally entangled states.

\subsection{Efficiency}
\subsubsection{fibers}
The downconverted modes are collected into AR coated singlemode fibers. The AR coating was done in situ by transferring a coating from a glass substrate to the fiber tip using an optically transparent epoxy (353ND). Singlemode fibers also allow us to mode-filter the downconverted light and carefully mode-match the pump and downconvered modes. This reduces the amount of stray light collected from unwanted modes and background or fluorescent light. 
\subsubsection{detectors}
We use Si APDs as single photon detectors. However their detection efficiency is low. To characterize the efficiency of our source we must account for the detector efficiency. This is of particular importance since there are now detectors with a near unit detection efficiency\cite{nam08}. To measure the detection efficiency of our Si APDs we used a scheme similar to \cite{migdall13}.  We compare the counts seen on the Si APDs to the expected number of photons from an attenuated 810nm laser. A bolometrically calibrated photodiode is used to calibrate a set of neutral density filters and a beamsplitter. We then use the calibrated photodiode to monitor the laser power while coupling the attenuated laser light to the Si APD. We measure a detection efficiency of $51.9\pm2.8\,\%$ and $46.7\pm2.5\,\%$ for the two primary detectors. 
\\subsubsection{The other large loss (splice)}
The source collection is optimized into singlemode 780-HP fibers. We then splice these fibers onto SMF28e fibers. The SMF28e fibers have a lower transmission and allow us to separate the entangled photon pairs by several meters with a negligible loss. However this splice between two different types of fibers with different core sizes and dopant concentrations has a few percent of loss. 
\subsubsection{Value uncorr}
Inclusive of this splicing loss and all optical and coupling losses, we obtain efficiencies of $39.16\pm0.16\,\%$ and $39.27\pm0.21\,\%$ efficiencies ($pairs/\sqrt{Singles_1Singles_2}$) in the two pump arms using Si APDs. 
\subsubsection{Apd correction}
Correcting only for the dark counts and detection efficiencies of these detectors, the source has an efficiency of $79.5\pm3\,\%$ and $79.8\pm3\,\%$ in the two pump arms. When the source produces an entangled state, the mode overlap between the two downconverted modes is still good and we have an entangled efficiency of $78.6\pm3\,\%$ {\bf I am still checking if i can improve this value}






\section{Conclusion}
For now this is just a list of the numbers i have

The efficiency of my detectors are $51.9\pm2.75\,\%$ and $46.7\pm2.5\,\%$ mentioned 

Left arm efficiency $= 39.16\pm0.16\,\%$  When corrected for detector loss this is $79.54\pm 3\,\%$ not mentioned together with R arm

Right arm efficiency $=39.27\pm0.21\,\%$ When corrected for detector loss this is $79.77\pm 3\,\%$

$\pm45\,^{\circ}$ basis entangled efficiency $=38.7\,\%   \pm ????$  When corrected for detector loss this is $78.61\pm3\,\%$ Said

For a final detection efficiency of $74\,\%$ optimal Bell violation will occur when $\theta=0.22272$. If I choose this state then the fidelity between the chosen state and the tomographically reconstructed state is $99.3\,\%  \pm 0.1$   The first bit is not mentioned

Pump bandwidth = ??????? 160MHz by data sheet mentioned

pairs/s/mW = 4069.5     {\bf needs to go somewhere}


pump waist is $265 \pm ????$ centered in the crystal both waists are mentioned

Collection waist is $160\,\mu m\pm????$

Bandwidth of downconverted by interferometer= ????


\section{What others have}


\end{document} 
